Author’s response to reviews

Title: Adherence to treatment and related factors among patients with chronic conditions in primary care: A cross-sectional study

Authors:

Cesar Fernandez-Lazaro (fernandezlazaro@usal.es)
Juan García-Gonzalez (jmgargon@upo.es)
David Adams (david.adam@point.edu)
Diego Fernandez-Lazaro (diego.fenandez.lazaro@uva.es)
Juan Mielgo-Ayuso (juanfrancisc.mielgo@uva.es)
Alberto Caballero-Garcia (albcaba@ah.uva.es)
Francisca Moreno Racionero (fmorenora@saludcastillayleon.es)
Alfredo Córdova (a.cordov@uva.es)
Jose Miron-Canelo (mirox@usal.es)

Version: 4 Date: 29 Aug 2019

Author’s response to reviews:

Editor Comments:

Comment:

1. Please clearly state in the Ethics approval section of the Declarations whether participants provided written or verbal consent.

Response:

We have now clearly stated that participants provided written consent to participate in the study in the Ethics Approval section (page 19, lines 417-418).
Comment:
2. Please also indicate the form of consent in the Consent for publication section.

Response:
We have now addressed this matter in the Consent for publication section (page 19, line 425).

Comment:
3. In the Availability of data and materials section, please indicate from whom data may be requested.

Response:
We have now addressed this matter in the Availability of data and materials section (page 19, line 427).

Comment:
4. Thank you for providing an authorship change form. Please remove it from the file inventory, as it is no longer needed at this stage of the editorial process.

Response:
We have removed the authorship change form from the inventory as requested.

Comment:
5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response:
We have uploaded all the documents as requested.